Author's response to reviews

Title: From disease to desire, pleasure to the pill: A qualitative study of adolescent learning about sexual health and sexuality in Chile.

Authors:

Anna K-J Macintyre (annakjmacintyre@gmail.com)
Adela R Montero (amontero@med.uchile.cl)
Mette Sagbakken (mette.sagbakken@hioa.no)

Version: 2 Date: 26 August 2015

Author's response to reviews:

The response to reviewers is provided in the attached cover letter.